Author's response to reviews

Title: Temporal trend and climate factors of hemorrhagic fever with renal syndrome epidemic in Shenyang City, China

Authors:

Xiaodong Liu (liuxd1983@126.com)
Baofa Jiang (bjiang@sdu.edu.cn)
Weidong Gu (weidonggu@cdc.gov)
Qiyong Liu (liuqiyong@icdc.cn)

Version: 9 Date: 1 December 2011

Author's response to reviews: see over
Dear Editors,

Thanks very much for your kind work and consideration on publication of our paper entitled “Temporal Trend and Climate Factors of Hemorrhagic Fever with Renal Syndrome Epidemic in Shenyang City, China” (ID: 2903271448538366). On behalf of my co-authors, we would like to express our great appreciation to editor and reviewers. Thank you and best regards.

We have checked the current version of our manuscript carefully and have made some minor corrections.

1. We cannot re-orientate the Table 1 to fit within a page, so we have moved it to the additional files.

2. We have removed the tracking from the manuscript.

Yours sincerely,

Xiaodong Liu